Author's response to reviews

Title: Facilitating adherence to physical activity: exercise professionals' experiences of the National Exercise Referral Scheme in Wales. A qualitative study

Authors:

Graham F Moore (MooreG@cardiff.ac.uk)
Laurence Moore (MooreL1@cardiff.ac.uk)
Simon Murphy (MurphyS7@cardiff.ac.uk)

Version: 4 Date: 15 December 2011

Author's response to reviews: see over
Dear Mr Dizon,

Thank you for returning our formatted manuscript for final editorial changes. Responses to editorial queries are listed below:

Q1: Please note that all tables must have at least 2 rows to be inserted as a table within this journal. In regards to Tables 2 and 3 please make the necessary changes so that they abide by journal standards for tables, or alternatively insert the information into the main body as text.
   Tables 2 and 3 have been edited as requested

Q2. Please provide complete bibliographic details of reference 9.
   These details have now been added

Q3. Please provide pages of reference 19.
   These details have now been added

I have also added details for reference 34, which at the time of submission was accepted, but has since been published online ahead of print.

I have proof read the manuscript, correcting a small number of typographical and grammatical errors.

Please feel free to contact me if you have any further queries.

Yours sincerely,

Dr Graham Moore